Author’s response to reviews

Title: Clinical value of DPOC for detecting and removing residual common bile duct stones (video)

Authors:

Jun-Jie Yang (junjie1001jjy@aliyun.com)
Xiong-Chang Liu (liuxiongchang1935@163.com)
Xiao-Qin Chen (chenxiaqin03cxq@sohu.com)
Qi-Yong Zhang (z297qiyong@sohu.com)
Tian-Rang Liu (tianrang197liu@sohu.com)

Version: 2 Date: 05 Jul 2019

Author’s response to reviews:

Comments 1 As indicated by our submission guidelines, please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.

Reply: Thank you. We have added the Conclusions heading after Discussion section.

Comments 2 Thank you for providing an Availability of data and materials statement for the Declarations section. Please note that while identifying/confidential patient data should not be published within the manuscript, a statement should be provided for where the de-identified raw patient data generated/analysed during your study can be found. If you wish to share the data through correspondence then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.” If you do not wish to share your data, please state this along with the reason.

Reply: Thank you. We have corrected the statement into “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request”.

Comments 3 Please kindly remove the first sentence found in your current Competing interests statement. For reference, the aforementioned sentence is as follow:

"All authors have contributed significantly to the manuscript and declare that the work is original and has not been submitted or published elsewhere."
In addition, please revise the Competing interests section so that it also addresses non-financial competing interests.

Reply: Thank you. We have changed the sentence into “The authors declare that they have no competing interests”.

Comments 4 Please note that all authors should be named by using their initials rather than full names in the Authors’ contributions section. For example, author John Smith should be represented as JS in this section.

Reply: Thank you. We have corrected the authors’ full names into their initials.

Comments 5 Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please state "Not applicable" in this section as no personal information is provided in your manuscript.

Reply: Thank you. We have corrected the statement into “Not applicable”.

Comments 6 Please revise the Objective heading in your Abstract section so that it appears as Background. Similarly, please revise the Introduction heading found in your main manuscript text so that it also appears as Background.

Reply: Thank you. We have corrected the section heading.

Comments 7 When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Reply: Yes. We have uploaded a clean copy.